Author’s response to reviews

Title: Choosing between Internet-based psychodynamic versus cognitive behavioral therapy for depression: a pilot preference study

Authors:

Robert Johansson (robert.johansson@liu.se)
Anna Nyblom (annfr624@student.liu.se)
Per Carlbring (per@carlbring.se)
Pim Cuijpers (p.cuijpers@psy.vu.nl)
Gerhard Andersson (gerhard.andersson@liu.se)

Version: 2 Date: 3 December 2012

Author’s response to reviews:

In the attached version of the manuscript, we have added the section "Ethics statement" which contains the text "The study was approved by the Regional Ethics Board of Linköping, Sweden (Registration number 2010/386-31). All participants provided signed informed consent via the online treatment platform."

By this, we have addressed the two comments provided by the editor, that is, both the comments regarding ethics and informed consent.

Please accept this manuscript as a candidate for publication in BMC Psychiatry. While the manuscript may seem appropriate for publication in BMC Psychology, we have communicated with the editor who has agreed that we submit to BMC Psychiatry as originally planned.